# Beyond the Classical Transport Laws of Electrochemistry: New Microscopic Approach to Ionic Conductance and Viscosity

Amalendu Chandra\*,<sup>†</sup> and Biman Bagchi\*,<sup>‡</sup>

Department of Chemistry, Indian Institute of Technology, Kanpur, India 208016, and Solid State and Structural Chemistry Unit, Indian Institute of Science, Bangalore, India 560 012

Received: March 21, 2000; In Final Form: June 14, 2000

The concentration dependence of the transport properties (i.e., the conductivity and the viscosity) of an electrolyte solution has been a subject of lively debate for a very long time. The foundation for understanding the transport properties of electrolyte solutions was laid down by Debye, Huckel, Onsager, and Falkenhagen who derived several limiting laws valid at low ion concentration. These classical laws have been rederived several times, although their extension to concentrated solutions has proven to be very difficult. We discuss a new microscopic approach toward understanding the transport laws of electrochemistry. This new approach is based on the general ideas of the mode coupling theory. We show that the mode coupling theory approach is appropriate in the present case because concentration effects arise from collective variables (like charge density and current) which are treated correctly by the mode coupling theory. The new theory can describe the crossover from the low to high concentration seamlessly. Our study yields microscopic expressions of both conductivity and viscosity in terms of static and dynamic structure factors of the charge and number densities of the electrolyte solution. The celebrated expressions of Debye, Huckel, and Onsager for static conductance, of Debye and Falkenhagen for frequency dependent electrolyte friction, and of Falkenhagen for the viscosity follow exactly from the present microscopic theory in the limit of very low ion concentration. Recently derived microscopic expressions of Chandra, Wei, and Patey for the frequency dependent conductivity can also be derived from the present scheme. The present theory is a self-consistent theory. For conductance, the agreement of the present theory with experimental results is satisfactory even up to one molar concentration. For viscosity, the theory seems to give the right trend and suggests directions for further improvement to explain the myriad of unexplained behavior known for a long time.

### 1. Introduction

The clasical transport laws and theories of electrochemistry have been widely applied to understand the effects of ion concentration on the diffusion of ions, ionic conductivity, and the viscosity of electrolytes.<sup>1–11</sup> The most celebrated among these is the Debye-Huckel-Onsager (DHO) law which predicts a square root concentration dependence of the ion conductance.12-13 There are two other laws which are often discussed in the literature. One of these is the Debye-Falkenhagen (DF) theory of the frequency dependence of ionic conductivity.<sup>14</sup> One often refers to Debye-Falkenhagen effect as the anomalous rise of conductivity with frequency at low frequencies which follows from the Debye-Falkenhagen theory. Another well-known theory is the Falkenhagen-Onsager-Fuoss (FOF) theory of the concentration dependence of the excess viscosity of ionic solutions.<sup>15,16</sup> This theory correctly explains the rise of viscosity with concentration in the limit of *very low* ion concentration.

The classical derivations of these transport laws were highly nontrivial, often involving astute use of electrohydrodynamics and irreversible thermodynamics—the classic 1932 article by Onsager and Fuoss<sup>16</sup> is a case in point. Not only have these

There have been several attempts to derive and extend the limiting transport laws to higher concentrations from microscopic point of view. Friedman<sup>18</sup> was the *first* to derive the

laws been hailed as the intellectual triumphs of the last century, they have also been tremendously successful in explaining concentration dependence of conductance at low ion concentration. However, these laws are certainly not perfect. They are valid only at low ion concentration and are applicable for strongly dissociative salt solutions such as NaCl and KCl in water. Both the successes and the limitations of these classical theories can be traced back to the basic conceptual framework on which they are based. This conceptual framework has two basic ingredients. First, the solvent is treated as a structureless continuum with a given static dielectric constant and viscosity. Second, there exists an ion atmosphere of net opposite charge around each ion due to ion attraction. The radius of this ion atmosphere is the Debye length ( $\lambda_D$ ) which has an inverse square root concentration dependence. The validity of the classical laws is crucially dependent on the value of the Debye length. When the Debye length is much larger than the molecular lengths (such as the radius of ions), then the classical description where the solvent is treated as a *dielectric continuum* is valid. The equilibrium theories based on this assumption are often referred to as ion attraction theory or the Debye-Huckel (DH) theory.<sup>17</sup>

<sup>&</sup>lt;sup>‡</sup> Solid State and Structural Chemistry Unit.

limiting law of conductance from the time correlation function formalism at microscopic level. The correlation function approach was also employed by Resibois<sup>4</sup> to derive the limiting law of conductance. The most notable among the studies at higher concentrations is the work of Friedman and co-workers<sup>19,20</sup> and of Blum and co-workers.<sup>21-23</sup> In the former approach, the motion of an ion is described as a Brownian particle in the force-field of other ions of the system. The forcefield is described in terms of ionic pair correlation functions. A dynamical theory at Smoluchowski level is used to calculate the transport properties at finite ion concentration. In the approach of Blum and co-workers, one uses the formal expressions of the transport properties derived earlier by Onsager and Fuoss<sup>16</sup> by using the continuity equation approach. These authors provided explicit expressions for the calculation of transport properties as functions of the equilibrium ionic pair correlation functions which are now available from the modern statistical mechanical theory of ionic solutions. Also, the difference between the self-diffusion and the conductivity was explicitly taken into account in the later work by Turg et al.<sup>22,23</sup> Expressions derived in both the approaches reduce to the DHO law in the limit of low concentration. Also, these theories do considerably better than the original DHO expression at higher concentrations. Much less effort has been devoted to understand the frequency dependence of conductance or the concentration dependence of viscosity.

Computer simulation is a complementary method to study transport phenomena in ionic solutions. Among the simulation techniques, the method of molecular dynamics (MD)<sup>24</sup> allows us to study the structural and dynamical properties of both ions and solvent molecules at Born-Oppenheimer level of description. This technique has been used to study self-diffusion and conductivity of simple model solutions such as ions in Stockmayer liquid<sup>25,26</sup> and also of more realistic solutions such as NaCl, KCl, and ZnBr<sub>2</sub> in water at finite concentrations.<sup>27-29</sup> However, the implementation of this technique for ionic solutions is computationally very costly because of the multiplicity of components, the long-range nature of ionic interactions, and the very long run that is required to obtain statistically meaningful averages. This is why a great majority of the simulation studies have used techniques of stochastic simulations.<sup>24</sup> In these methods, the solution can be treated at the McMillan-Mayer level of description<sup>9</sup> where solvent particles are not considered explicitly: they are represented by a dielectric continuum and the solute particles interact through solvent averaged potentials. Brownian Dynamics (BD) and Langevin Dynamics (LD) are the two most often used stochastic simulation methods that have been employed to calculate the selfdiffusion and conductance of various aqueous ionic solutions.<sup>30-32</sup> Reasonably good agreement has been found with experimental results for diffusion and conductance. The study of the frequency dependence of conductivity has also been carried out for model ionic solutions.<sup>25,26,33</sup> These studies also showed the breakdown of Debye-Huckel and Debye-Huckel-Onsager classical laws at finite concentrations. To the best of our knowledge, no simulation study has yet been carried out to calculate the viscosity of ionic solutions.

Certainly one valuable lesson to be learned from the above studies of diffusion and conductance is that the dielectric continuum model itself may be trusted up to concentration as high as 1 M solution; the original DHO law breaks down even at 0.01 M. However, this relative success of the dielectric continuum model may be limited only to the calculation of diffusion and conductance and might not be extendable to other transport properties, like the viscosity. In addition, there are several aspects of the earlier theoretical approaches which require further improvement. Most importantly, these theories are not self-consistent—a limitation which must be removed before one can treat higher concentrations.

Recently, we have been able to derive all the three abovementioned classical transport laws using the basic concepts of mode coupling theory and the time dependent density functional theory.<sup>34–37</sup> The resulting expression for the ionic conductance involves the dynamic structure factors and the current-current correlation functions of the ions. In fact, the ion atmosphere term is shown to correspond to the relaxation of dynamic structure factors of the ions and the electrophoretic term to the time correlation function involving the charge density and the current terms. When microscopic expressions are evaluated, agreement (at least) at par with the earlier theories are found for the electrolyte conductance. This approach also provides microscopic expressions for the frequency dependence of ionic conductivity and for the ionic contribution to the viscosity of an electrolyte solution.

The objective of this Article is to summarize and discuss the mode coupling theory approach to the transport properties of electrolyte solutions. The organization of the rest of the Article is as follows. In the next section, we discuss the elements of the mode coupling theory which shall be used here. Section 3 contains discussion of the zero frequency conductance and derivation of the DHO law. Section 4 contains the discussion of the frequency dependent conductivity and the Debye–Falkenhagen effect. Section 5 summarizes the work on the concentration dependence of viscosity. Numerical results of the electrolyte friction, conductivity and viscosity are discussed in section 6. Section 7 presents a discussion of a full microscopic theory. Section 8 concludes with a discussion of the future problems.

#### 2. Mode Coupling Theory Approach to Electrochemistry

The mode coupling theory (MCT) was originally developed to explain the dramatic decrease in the values of the transport properties near the critical point. More recently, it has been used (with mixed success) to describe the anomalous dynamics in the supercooled liquid near the glass transition.<sup>38,39</sup> While the basic idea behind the latter development is somewhat different from the former one, the structure of the theory has remained essentially the same. MCT is the natural framework to use for describing the transport phenomena of electrolyte solutions because here one is interested in the effects of collective dynamics such as ion atmosphere relaxation on the single particle properties. For applications to electrolytes, one needs to identify the slow variables which control the dynamics. In an electrolyte solution at not too high concentration, these slow variables are the charge density and the ion current. It is the charge density and the current which give rise to the square root concentration dependence of the conductance and the viscosity through ionic pair correlation functions at low concentration. Actually, the relevant length in the problem is still the Debye length whose inverse goes to zero as the square root of concentration of the ions.

At low concentration, ionic conductance is essentially driven by the self-diffusion of the ions. The latter is a single particle property in the sense that this property can be measured by following the motion of a single tagged ion. However, this single particle motion is intimately connected to the collective, manybody dynamics of the surrounding liquid and this coupling between the collective and the single particle motion makes the calculation of self-diffusion coefficient of a tagged solute an extremely difficult problem.<sup>40</sup> One traditional approach has been to relate the friction on the ion to the viscosity of the medium, combined with the use of the exact Einstein relation between diffusion and friction.<sup>38</sup> The radius of the ion has often been used as a fit parameter. Although this method had some numerical success, the use of hydrodynamics for small particles has been shown to be seriously flawed on fundamental ground.<sup>41</sup>

The mode coupling theory provides a systematic way to treat this coupling between the single particle and the collective dynamics. There are several different approaches to this theory and all of them lead essentially to the same equations. The main idea is to express the friction on the tagged molecule (here ion) in terms of the relevant time correlation functions. For example, diffusion of a molecule should be coupled to density fluctuations which are responsible for relaxation of the surrounding cage and also to the current modes which contribute to the random motion of the molecule via the natural currents present in the liquid. Under some general approximations, the diffusion of a tagged molecule can be written as  $D = D_{\rm mic} + D_{\rm hyd}^{42}$  where  $D_{\rm mic}$  denotes the microscopic contribution to diffusion which consists of a bare term and the collective density term and  $D_{hvd}$ is the hydrodynamic part which contains the contribution from the current modes.

The mode coupling theory has been constructed here considering only the collective ionic modes as the slow variables. This approach is valid at low ion concentration where the solvent modes relax much faster than the ionic modes. An extension of this approach has been discussed in section 7.

The hall mark of any mode coupling theory is a self-consistent calculation of the transport properties and the time correlation functions. For example, the contribution of the density term to diffusion itself depends on the self-diffusion coefficient. One eventually has a "mode coupling necklace" where all the terms are connected to each other, often by nonlinear equations. These equations are to be solved self-consistently. In the present work, MCT will be used to calculate the frequency dependent ionic friction and the self-diffusion coefficients. From the self-diffusion coefficients, we will calculate the conductance and the frequency dependent conductivity. As we will see in section 5, MCT also provides an expression for the ionic contribution to the viscosity in terms of the static and dynamic charge density structure factors of the solution.

# **3.** The Total Electrolyte Friction and Limiting Ionic Conductance

In the limit of very low but finite ion concentration, the conductance is described by the well-known Debye–Huckel– Onsager limiting law<sup>12,13</sup>

$$\Lambda_{\alpha}(c_{\alpha}) = \Lambda_{\alpha}^{0} - \left[\mathcal{A} + \mathcal{B}\Lambda_{\alpha}^{0}\right]\sqrt{c_{\alpha}} \tag{1}$$

where  $\Lambda_{\alpha}$  is the conductance of the ionic species  $\alpha$  at concentration  $c_{\alpha}$  and  $\Lambda_{\alpha}^{0}$  is the conductance at infinite dilution which is obtained by extrapolating the concentration to zero. A and B are the two constants which are determined by the properties of the ions and the medium.

In this section, we will calculate the conductance from a microscopic calculation of the electrolyte friction. We start with the following generalized Langevin equation for time evolution of the velocity of a single tagged ion of charge  $q_s^{38}$ 

$$\frac{\partial}{\partial t}v_{\rm s}(t) = -\int_0^\infty {\rm d}t\zeta_{\rm s}(t-t')v_{\rm s}(t') + f_{\rm s}(t) \tag{2}$$

where  $\zeta_s(t)$  is the total friction acting on the single tagged ion and  $f_s(t)$  is the so-called random force. The Laplace transform of  $\zeta_s(t)$  is  $\zeta_s(\omega)$  where  $-i\omega$  is the Laplace frequency variable and *i* is the imaginary number. The frequency dependent selfdiffusion coefficient  $D_s(\omega)$  is related to the friction  $\zeta_s(\omega)$  by the following generalized Nernst–Einstein relation

$$D_{\rm s}(\omega) = \frac{k_{\rm B}T}{m(-i\omega + \zeta_{\rm s}(\omega))} \tag{3}$$

where  $k_{\rm B}$  is Boltzmann constant, T is the temperature, and m is the mass of the tagged ion. As discussed in the last section, the total friction on the tagged ion can be decomposed into two parts: A microscopic relaxation term  $\zeta_{s;mic}(\omega)$  and a hydrodynamic term  $\zeta_{s;hyd}(\omega)$ . The inverses of these two quntities add to give the inverse of the total ionic friction. The microscopic friction is most easily analyzed by using the Kirkwood's formula for friction which expresses it in terms of an integration over the force-force time correlation function.<sup>43</sup> Since the time dependent force on the tagged ion has contributions from solvent density and polarization fluctuations<sup>44-46</sup> and also from ion atmosphere fluctuations, one can decompose the total microscopic friction into a solvent contribution  $\zeta_{s,mic}(c=0)$  which is assumed to be a constant independent of ion concentration and a concentration dependent ion contribution  $\delta \zeta_{s,mic}(\omega)$  which originates from interaction with the ion atmosphere fluctuations. Here we calculate the ion contribution to the microscopic friction. Thus, the solvent contribution is not calculated microscopically, it determines the ion diffusion at infinite dilution whose value is assumed to be known from experiments.

The hydrodynamic contribution originates from the coupling of the ion velocity to the relevant current modes of the solution. Mode coupling theory directly provides an expression of the contribution of the currents of the system to the diffusion coefficient, that is, to the inverse friction.<sup>38,47-48</sup> Since the current modes of the solution consists of both the solvent and ion currents, both contribute to the inverse of the friction. The hydrodynamic contribution to the total friction can, therefore, be expressed as<sup>34,35</sup>  $\zeta_{s;hyd}(z)^{-1} = \zeta_{s;hyd}(c=0)^{-1} + \delta \zeta_{s,hyd}(z)^{-1}$ where  $\zeta_{s;hyd}(c=0)$  is the usual viscous friction which now also includes the additional dielectric friction contribution due to the polarization current.<sup>49,50</sup>  $\delta \zeta_{s,hyd}(\omega)$  represents the electrolyte friction on the tagged ion due to coupling with the ion atmosphere current. The latter is commonly known as the electrophoretic effect. As before, the solvent contribution is not calculated in the present work. It constitutes part of the ion diffusion at infinite dilution which is assumed to be known from experiments. The ion contribution to the hydrodynamic part or the electrophoretic effect is calculated here microscopically by using the mode coupling theory.

## **3.1.** The Calculation of the Ion Atmosphere Contribution to the Microscopic Friction: The Asymmetric Effect

The ion atmosphere contribution to the microscopic electrolyte friction can be calculated from the correlation of the force exerted by the fluctuating ion atmosphere on the tagged ion by using the Kirkwood's formula.<sup>43</sup> In time domain, the microscopic electrolyte friction can be expressed as

$$\delta \zeta_{\text{s,mic}}(t) = \frac{1}{3k_{\text{B}}TV} \int d\mathbf{r} \left\langle \mathbf{F}_{\text{s}}(\mathbf{r},t) \cdot \mathbf{F}_{\text{s}}(\mathbf{r},0) \right\rangle$$
(4)

where  $\mathbf{F}_{s}(\mathbf{r},t)$  is the force exerted on the tagged ion due to its interaction with all other ions in the solution and *V* is the volume of the system. An expression for  $\mathbf{F}(\mathbf{r},t)$  can be obtained from

density functional theory and it is given by<sup>34,35</sup>

$$\mathbf{F}(\mathbf{r},t) = k_{\rm B} T n_{\rm s}(\mathbf{r},t) \nabla \sum_{\alpha} \int d\mathbf{r}' \, c_{\rm s\alpha}(\mathbf{r},\mathbf{r}') \delta \rho_{\alpha}(\mathbf{r}',t) \quad (5)$$

where  $n_{\rm s}(\mathbf{r},t)$  represents the position ( $\mathbf{r}$ ) and time (t) dependent density distribution of the tagged ion and  $\delta \rho_{\alpha}(\mathbf{r},t)$  is the collective density fluctuation which is equal to  $\rho_{\alpha}(\mathbf{r},t) - \rho_{\alpha}$ , where  $\rho_{\alpha}$  is the average number density of ionic species  $\alpha$  and  $c_{\rm s\alpha}(\mathbf{r},\mathbf{r}')$  is the direct correlation function between species  $\alpha$  and the tagged ion. Because of the presence of convolution integral in eq 5 and of isotropy of the system, final expressions are much simpler in the Fourier space. The final expression for the microscopic electrolyte friction has the following simple expression<sup>34,35</sup>

$$\delta \zeta_{\text{s,mic}}(t) = \frac{k_{\text{B}}T}{3(2\pi)^3} \sum_{\alpha,\beta} \int d\mathbf{k} k^2 c_{s\alpha}(k) \sqrt{\rho_{\alpha}\rho_{\beta}} G_{\alpha\beta}(k,t) c_{s\beta}(k) F_{s}(k,t)$$
(6)

where the wavevector **k** is the Fourier variable conjugate to **r** and  $G_{\alpha\beta}(k,t)$  is the Fourier transform of the ionic van Hove function between species  $\alpha$  and  $\beta$  which is defined as

$$G_{\alpha\beta}(k,t) = (N_{\alpha}N_{\beta})^{-1/2} \left\langle \rho_{\alpha}(\mathbf{k},t)\rho_{\beta}(-\mathbf{k})\right\rangle$$
(7)

where  $\langle \cdots \rangle$  denotes average over an equilibrium ensemble and  $\rho_{\alpha}(\mathbf{k},t)$  is the Fourier transform of  $\rho_{\alpha}(\mathbf{r},t)$ .  $N_{\alpha}$  and  $N_{\beta}$  are the number of ions of species  $\alpha$  and  $\beta$  in the solution, respectively. Later, we will denote the Fourier-Laplace transform of the van Hove function as  $G_{\alpha\beta}(k,\omega)$ .  $F_{s}(k,t)$  is the self-dynamic structure factor of the tagged ion. Clearly, the effects of self-motion on the ionic friction is contained in  $F_{s}(k,t)$ . Equation 6 is an important result. Note that the same expression can be defined from a more rigorous analysis, as the one given by Sjogren and Sjolander.<sup>47</sup>

A microscopic calculation of the density relaxation term requires tractable expressions of the wavenumber (k) and time (t) (or frequency  $(\omega)$ ) dependent ionic van Hove functions. Such expressions can be obtained from the time dependent density functional theory. In the present work, we ignore the explicit dynamical coupling of ions with the solvent and we include the effects of the solvent in an effective manner. Thus, for the calculation of the ion-ion correlations, it is assumed that the solvent is a dielectric continuum with dielectric constant  $\epsilon$  and the ions interact with a solvent averaged interaction potential, i.e., with a long-range Coulomb potential scaled by the dielectric constant of the solvent.<sup>38</sup> Also, the dynamics of ions is now described by generalized Smoluchowski theory with appropriate solvent averaged mean potential and diffusion coefficients. We note that the assumption of solvent continuum for the calculation of the ion-ion correlations is expected to be valid in the limit of very low ion concentrations. Use of time dependent density functional theory leads to the following generalized Smoluchowski equation for the frequency dependent van Hove function35,51

$$G_{\alpha\beta}(k,\omega) = \left[-\omega + D_{\alpha}(\omega)k^{2}\right]^{-1}S_{\alpha\beta}(k) + \frac{D_{\alpha}(\omega)k^{2}}{-i\omega + D_{\alpha}(\omega)k^{2}}\sum_{\gamma=1}^{2}\sqrt{\rho_{\alpha}\rho_{\gamma}}c_{\alpha\gamma}(k)G_{\gamma\beta}(k,\omega)$$
(8)

where the  $D_{\alpha}(\omega)$  is the frequency dependent diffusion coefficient of species  $\alpha$  and  $c_{\alpha\beta}(k)$  is the Fourier transform of the direct correlation function between species  $\alpha$  and  $\beta$ .  $S_{\alpha\beta}(k) = G_{\alpha\beta}(k,t) = 0$  where  $S_{\alpha\beta}(k)$  is the partial static structure factor between species  $\alpha$  and  $\beta$ .

The coupled equations as given by eq 8 for the van Hove functions can be solved to obtain the following explicit result for the time dependence of the ionic van Hove functions<sup>35,51</sup>

$$\sum_{\alpha\beta} \sqrt{\rho_{\alpha}\rho_{\beta}} q_{\alpha}q_{\beta}G_{\alpha\beta}(k,\omega) = \frac{1}{Z(k,\omega)} [-i\omega\sum_{\alpha\beta} \sqrt{\rho_{\alpha}\rho_{\beta}} q_{\alpha}q_{\beta}S_{\alpha\beta}(k) + D_{2}(\omega)k^{2}\Delta(k)\{\sqrt{\rho_{1}}q_{1}S_{11}(k) + \sqrt{\rho_{2}}q_{2}S_{12}(k)\}^{2} + D_{1}(\omega)k^{2}\Delta(k)\{\sqrt{\rho_{2}}q_{2}S_{22}(k) + \sqrt{\rho_{1}}q_{1}S_{21}(k)\}^{2}]$$
(9)

where

$$Z(k,\omega) = -\omega^2 - i\omega\Delta(k)[D_1(\omega)k^2S_{22}(k) + D_2(\omega)k^2S_{11}(k)] + D_1(\omega)D_2(\omega)k^4\Delta(k)$$
(10)

and  $\Delta(k) = [S_{11}(k)S_{22}(k) - S_{12}(k)^2]^{-1}$ . The Laplace transform of the self-dynamic structure factor of the tagged ion can be described by the following equation

$$F_{\rm s}(k,\omega) = \frac{1}{-i\omega + D_{\rm s}(\omega)k^2} \tag{11}$$

The analytical solutions of the ionic structure factors and the direct correlation functions are known in the literature. Thus, eqs 6-10, together with the hydrodynamic term discussed later, form a set of self-consistent equations which can be solved to obtain the microscopic electrolyte friction.

**3.2.** Calculation of the Ion Atmosphere Contribution to the Hydrodynamic Friction: The Electrophoretic Effect. The ion atmosphere contribution to the hydrodynamic friction originates from the coupling of the velocity of the tagged ion with the current mode of the ions. A formal expression of this friction can be derived from mode coupling theory by using  $\rho^{c}(k)\mathbf{j}(-k)$  as the relevant binary product where  $\rho^{c}(k)$  and  $\mathbf{j}(k)$  are the charge density and total ion current of the solution, respectively. The hydrodynamic friction is then given by

$$\frac{k_{\rm B}T}{\delta \zeta_{\rm hyd}(t)} = \sum_{k} \sum_{k'} \langle u_{0x}, \rho^{\rm c}(\mathbf{k})\mathbf{j}(-\mathbf{k})\rangle \langle \rho^{\rm c}(\mathbf{k})\mathbf{j}(-\mathbf{k}), \rho^{\rm c}(\mathbf{k}) \times \mathbf{j}(-\mathbf{k})\rangle^{-1} \times \langle \rho^{\rm c}(\mathbf{k})\mathbf{j}(-\mathbf{k}), e^{i\Delta t} \rho^{\rm c}(\mathbf{k}')\mathbf{j}(-\mathbf{k}')\rangle \langle \rho^{\rm c}(\mathbf{k}')\mathbf{j}(-\mathbf{k}'), \rho^{\rm c}(\mathbf{k}')\mathbf{j}(-\mathbf{k}')\rangle^{-1} \times \langle \rho^{\rm c}(\mathbf{k}')\mathbf{j}(-\mathbf{k}'), u_{0x}\rangle$$
(12)

where  $u_{0x}$  is the velocity of the tagged ion along a particular direction (say, *x*) and *L* is the Liouville operator. The vertexes and the time correlation function in the above expression can be evaluated by using the following decomposition of the charge density and the ion current  $\rho^{c}(k) = \rho_{0}^{c}(k) + \rho_{1}^{c}(k) + \rho_{2}^{c}(k)$  and  $\mathbf{j}(k) = \mathbf{j}_{0}(k) + \mathbf{j}_{1}(k) + \mathbf{j}_{2}(k)$  where  $\rho_{0}^{c}(k)$  and  $\mathbf{j}_{0}(k)$  are the charge density and current of the tagged ion and  $\rho_{\alpha}^{c}(k)$  and  $\mathbf{j}_{\alpha}(k)$  are those of species  $\alpha$  surrounding the tagged ion.

After a strightforward analysis, the following expression is obtained for the electrophoretic friction<sup>34,35</sup>

$$\frac{k_{\rm B}T}{\delta\zeta_{\rm s,hyd}} = \frac{1}{3\pi^2 \rho N} \int_0^\infty dt \int_0^\infty dk k^2 [\rho_1 G_{11}^{\rm d}(k,t) - \rho_2 G_{12}^{\rm d}(k,t)] C^{\rm T}(k,t)$$
(13)

where  $\rho$  is the total ion density and *N* is the total number of ions in the solution.  $G^{d}_{\alpha\beta}(k,t)$  is the distinct part of the van Hove function between species  $\alpha$  and  $\beta$  and  $C^{T}(k,t)$  is the transverse part of the current correlation function. Note that the relaxation of the transverse velocity correlation function occurs much faster than the distinct van Hove functions and the major contribution to the above integral comes from the short time region where  $G^{d}_{11}(k,t) \approx h_{11}(k)$  and  $G^{d}_{12}(k,t) \approx h_{12}(k)$  where  $h_{\alpha\beta}$ -(*k*) is the Fourier transform of the total pair correlation function between species  $\alpha$  and  $\beta$ . We also assume an exponential decay of  $C^{T}(k,t)$  of the following form<sup>42</sup>

$$C^{\mathrm{T}}(k,t) = \frac{Nk_{\mathrm{B}}T}{m} \mathrm{e}^{-\eta k^{2}t/\rho m} \tag{14}$$

where  $\eta$  is the viscosity of the solution. Equation 14 can be substituted in eq 13 to obtain an expression for the electrophoretic friction in terms of viscosity  $\eta$  and ionic pair correlation functions.

Note that both  $F_s(k,t)$  and  $G_{\alpha\beta}(k,t)$  depend on the self-diffusion coefficient of the ions. On the other hand, these correlation functions determine the self-diffusion coefficient through eqs 6-11 and 13. Thus, the calculation of *D* requires a selfconsistent solution of these expressions. This is nontrivial. As already mentioned, such self-consistent calculation is the hallmark of any MCT calculation. In fact, strictly speaking, one should include the expression for viscosity (derived in section 5 below) in this self-consistent calculation. We have neglected this complication here because the concentration dependence of viscosity is not significant at low and intermediate concentrations.

**3.3. Reduction to Debye–Huckel–Onsager Limiting Law.** It is shown here how the present molecular theory reduces to the well-known DHO law when applied to solutions of very low concentration. For convenience, the expression for the microsocpic friction is rewritten in the following form<sup>35</sup>

$$\delta \zeta_{\text{s,mic}} = \frac{k_{\text{B}}T}{3(2\pi)^3} \int_0^\infty dt \int d\mathbf{k} k^2 [h_{\text{s}}(k)] [S(k)]^{-1} [G(k,t)] [S(k)]^{-1} [G(k,t)] [S(k)]^{-1} [h_{\text{s}}(k)]^{\dagger} F_{\text{s}}(k,t)$$
(15)

where  $[h_s(k)]$  is the row matrix  $[\sqrt{\rho_1}h_{s1}(k)\sqrt{\rho_2}h_{s2}(k)]$ ,  $[h_s(k)]^{\dagger}$  is the transpose of  $[h_s(k)]$ . [S(k)] and [G(k,t)] are the 2 × 2 structure factor and van Hove function matrices of the ions.

3.3.1. Debye-Huckel Expression. We shall first derive the original Debye-Huckel expression<sup>12</sup> for the asymmetric effect. For this case, we ignore the self-motion, i.e., we set  $F_s(k,t) = 1$  and we assume that the diffusion coefficients of the positive and negative ions are equal i.e., we assume  $D_s = D_1 = D_2$ . In this simplified case, the time integral of the van Hove function is given by

$$\int_0^\infty dt [G(k,t)] = \frac{[S(k)]}{k^2 D_s [S(k)]^{-1}}$$
(16)

We substitute eq 16 in eq 15 to obtain the following expression for the microscopic friction

$$\delta \zeta_{\text{s,mic}} = \frac{k_{\text{B}}T}{3(2\pi)^3} \frac{1}{D_s} \int d\mathbf{k} [h_{\text{s}}(k)] [h_{\text{s}}(k)]^{\dagger}$$
(17)

Since our goal is to derive the DHO law, we assume the ions to be point ions and use the following Debye–Huckel solution<sup>17</sup>

for the Fourier transform of the pair correlation function

$$h_{\alpha\beta}(k) = -\frac{4\pi q_{\alpha}q_{\beta}}{\epsilon k_{\rm B}T} \frac{1}{k^2 + \kappa_{\rm D}^2}$$
(18)

in eq 17 and evaluate the integral over the wavevector space to obtain the following expression for  $\delta \zeta_{s,mic}$ 

$$\delta \zeta_{\rm s,mic} = \frac{q_{\rm s}^2 \kappa_{\rm D}}{6\epsilon D_{\rm s}} \tag{19}$$

where  $\kappa_D$  is the inverse Debye screening length given by

$$\kappa_{\rm D} = \left(\frac{4\pi}{\epsilon k_{\rm B}T} \sum_{\alpha} \rho_{\alpha} q_{\alpha}^2\right)^{1/2} \tag{20}$$

The above expression is precisely the one originally derived by Debye and Huckel ignoring self-motion of the tagged ion.<sup>12</sup>

3.3.2. Onsager's Correction: DHO Expression. Next, we derive Onsager's correction<sup>13</sup> by including the self-motion but still assume that the diffusion coefficients are equal. The microscopic friction can now be written as

$$\delta \zeta_{\text{s,mic}} = \frac{k_{\text{B}}T}{3(2\pi)^3} \int_0^\infty dt \int d\mathbf{k} k^2 [h_{\text{s}}(k)] [S(k)]^{-1} [G'(k,t)] \times [S(k)]^{-1} [h_{\text{s}}(k)]^{\dagger} F_{\text{s}}(k,t)$$
(21)

where [G'(k,t)] is the modified van Hove function matrix given by  $[G'(k,t)] = e^{-D_kk^2t}[G(k,t)]$ . The time integral of the modified van Hove matrix can be derived from the molecular hydrodynamic equations described earlier and is given by

$$\int_{0}^{\infty} dt [G'(k,t)] = \frac{[S(k)]}{D_{s}k^{2} + D_{s}k^{2}[S(k)]^{-1}}$$
(22)

We substitute eq 22 in eq 21 and use the relation between the structure factor and the pair correlation function matrices to obtain

$$\delta \zeta_{s,mic} = \frac{k_{\rm B}T}{6D_{\rm s}(2\pi)^3} \int d\mathbf{k} [h_{\rm s}(k)] \times \left\{ 1 - \frac{1}{2} [h(k)] + \frac{1}{4} [h(k)]^2 ... \right\} [h_{\rm s}(k)]^{\dagger}$$
(23)

where [h(k)] is the 2 × 2 matrix of ion—ion pair correlation functions. The various terms in the right-hand side of the above equation can be evaluated analytically by using Debye—Huckel solution for the ionic pair correlation functions, and the final result is<sup>35</sup>

$$\delta \zeta_{\text{s,mic}} = \frac{q_{\text{s}}^2 \kappa_{\text{D}}}{6\epsilon D_{\text{s}}} (2 - \sqrt{2}) \tag{24}$$

The above expression is identical with the one derived by Onsager<sup>13</sup> for a binary electrolyte by including the self-Brownian motion of the tagged ion. We note that the quantity  $(2 - \sqrt{2})$  is denoted as the *w* factor in DHO law for a binary electrolyte.

3.3.3. The Generalized DHO Expression. Now we consider the more general case where self-motion is included and the diffusion coefficients of the positive and negative ions are taken to be unequal, i.e.,  $D_1 \neq D_2$ . In this case, the time integral of the modified van Hove function matrix is given by

$$\int_0^\infty dt [G'(k,t)] = \frac{[S(k)]}{k^2 [D_s] + k^2 [D] [S(k)]^{-1}}$$
(25)

where

$$[D_{\rm s}] = \begin{bmatrix} D_{\rm s} & 0\\ 0 & D_{\rm s} \end{bmatrix}$$

and

$$[D] = \begin{bmatrix} D_1 & 0 \\ 0 & D_2 \end{bmatrix}$$

Again, we substitute eq 25 in eq 21, use the relation between the structure factor and the pair correlation function and evaluate the integral over wavevector space to obtain the following final expression for the microscopic friction

$$\delta \zeta_{\text{s,mic}} = \frac{q_{\text{s}}^2 \kappa_{\text{D}}}{6\epsilon D_{\text{s}}^2} 2 \left[ 1 - \left( 1 - \frac{\sum_{\alpha} \rho_{\alpha} q_{\alpha}^2 D_{\text{s}} / (D_{\text{s}} + D_{\alpha})}{\sum_{\alpha} \rho_{\alpha} q_{\alpha}^2} \right)^{1/2} \right] \quad (26)$$

The above expression reduces to the DHO expression for electrolyte friction when  $D_s$  is taken to be equal to  $D_1$  and  $D_2$ . Also, the above expression becomes identical with the one of Onsager (derived in 1945)<sup>52</sup> and of Friedman and co-workers<sup>19</sup> when  $D_s$  and  $D_\alpha$  ( $\alpha = 1,2$ ) terms in the right-hand side of the above equation are replaced  $D_s^0$  and  $D_\alpha^0$  which are the values of the ion diffusion coefficients in the limit of infinite dilution.

Now we discuss the calculation of the hydrodynamic term in the limit of zero ion size and very low concentration. In this limiting situation, the Debye–Huckel solution of the ionic pair correlation functions can be used. We also set  $\sigma = 0$  in eq 13 and we evaluate the integral over the wavevector space to obtain

$$\frac{1}{\zeta_{\text{s,hyd}}} = \frac{1}{\zeta_{\text{s,hyd}}(c=0)} - \frac{\kappa_{\text{D}}}{6\pi\eta}$$
(27)

where the second term represents the electrophoretic contribution to the total diffusion. We note that the above expression for the electrophoretic contribution is a general result which does not depend on any hydrodynamic boundary condition. This is exactly what was pointed out by Onsager.<sup>13</sup>

Equations 26–27, together with the relation between the diffusion coefficient and the conductance<sup>53</sup> and between  $c_{\alpha}$  (the concentration of  $\alpha$ th species in moles per liter) and  $\rho_{\alpha}$  provide the following general equation for the conductance at very low concentration

$$\Lambda_{\alpha}(c_{\alpha}) = \Lambda_{\alpha}^{0} - \left[\frac{\sqrt{2\pi N_{0}}q_{\alpha}^{2}F}{3\pi\eta(1000\epsilon k_{\rm B}T)^{1/2}} + \frac{\sqrt{2\pi N_{0}}q_{\alpha}^{2}|q_{\alpha}|w_{\alpha}}{3\sqrt{1000}(\epsilon k_{\rm B}T)^{3/2}}\Lambda_{\alpha}^{0}\right]\sqrt{c_{\alpha}} (28)$$

where

$$w_{\alpha} = 2 \left[ 1 - \left( \frac{1}{2} \sum_{\beta} \frac{\Lambda_{\alpha}}{\Lambda_{\alpha} + \Lambda_{\beta}} \right)^{1/2} \right]$$
(29)

where  $N_0$  is Avogadro number and F is Faraday. The above equations are to be solved self-consistently, in general, to obtain

the numerical values of the conductances of positive and negative ions. Note that the factor  $w_{\alpha}$  goes over exactly to the DHO value  $(2 - \sqrt{2})$  for a binary electrolyte when one assumes  $\Lambda_1 = \Lambda_2$ . More importantly, in the same limit eq 28 becomes identical with the DHO limiting law. Furthermore, the above equation reduces to the expression of Friedman and coworkers<sup>19</sup> when one replaces  $\Lambda_{\alpha}$  terms in the right-hand side of the equation by  $\Lambda_{\alpha}^0$ . This is exact to the order  $\sqrt{c_{\alpha}}$ . Equation 29 predicts different values of  $w_{\alpha}$  for positive and negative ions when their mobilities are different. The original DHO law does not reflect this asymmetry although it was later noticed by Onsager.<sup>52</sup>

### 4. Frequency Dependent Ionic Conductivity: Debye-Falkenhagen Effect

The frequency dependent specific conductivity  $\sigma(\omega)$  describes the motion of ions in the presence of a time dependent extenal field, where  $\omega$  is the oscillation frequency of the field. When an ion moves in electrolyte solution, the atmosphere cannot immediately follow the motion of the central ion and becomes asymmetric causing a retarding effect on the motion of the ion. At zero frequency, this relaxation effect leads to the  $B\Lambda_0\sqrt{c}$ term in eq 1. In the presence of an oscillating field, the central ion oscillates and the ion atmosphere gets less time to relax and cannot follow the oscillations of the ion. The average effect is that the ion atmosphere remains more symmetric (or less asymmetric) compared to that for a moving ion in a static field. As a result, the effects of the asymmetry of the ion atmosphere is reduced causing a net reduction of the electrolyte friction and an enhancement of the conductivity at low frequency. This "anomalous" increase of conductivity at low frequency is known as the Debye-Falkenhagen effect.<sup>14</sup> At high frequency, the conductivity decreases because the ions oscillate so fast that the net ionic motion along a particular direction is smaller than that in the presence of a static or low-frequency field. By using a diffusion equation approach for the time dependence of ion atmosphere, Debye-Falkenhagen derived the following rather unusual looking expression for the frequency dependent electrolyte friction<sup>14</sup>

$$\xi_{\rm DF}(\omega) = \xi_{\rm DF}(0) \frac{1 + \sqrt{q}}{1 + [q(1 - i\omega\tau_{\rm atm})]^{1/2}}$$
(30)

where  $\zeta_{\rm DF}(0)$  is the zero-frequency friction, q = 1/2 for a symmetric binary electrolyte and  $\tau_{\rm atm}$  is the relaxation time of the ion atmosphere given by

$$\tau_{\rm atm} = \frac{1}{(D_1 + D_2)q\kappa_{\rm D}^2}$$
(31)

Here,  $D_1$  and  $D_2$  are the self-diffusion coefficients of the positive and negative ions, respectively, and  $\kappa_D$  is the inverse Debye screening length defined by eq 20.

As the DF expression is based on Debye–Huckel theory, it is valid only at very low ion concentrations. For a typical 0.001 M solution of a 1:1 salt,  $\tau_{atm} \sim 10^{-7}$ s, and therefore, the dispersion of the DF friction for such a solution is predicted to occur in the MHz region. The electrophoretic force, on the other hand, responds at rates comparable to that of molecular velocity correlations. The velocity correlation times in solutions are of the order of 0.1 ps, and thus, the dispersion of the electrophoretic contribution occurs at a frequency much higher than GHz. In DF theory, the frequency dependence of the electrophoretic effect is not considered. Thus, the well-known *Debye–Falk-enhagen effect* of increasing conductivity with frequency arises solely from the decrease of the friction from the ion atmosphere.

Actually, the Debye–Falkenhagen effect seems to have never been confirmed satisfactorily. In a new twist to this old problem, Anderson, in a paper entitled "The Debye–Falkenhagen Effect: Experimental Fact or Fiction?",<sup>54</sup> has recently questioned the original analysis of Falkenhagen. The difficulty of observing this effect is that the effect is predicted at very small concentration and also the magnitude of the effect is rather small.

In the following we discuss the recent mode coupling theory analysis of this problem. We calculate the frequency dependent electrolyte friction and the ionic diffusion coefficients by using the same theory discussed in the last section. It is assumed that the frequency dependent conductivity  $\sigma(\omega)$  is related to the frequency dependent ionic diffusion coefficients  $D_{\alpha}(\omega)$  by the following generalized Nernst–Einstein relation<sup>36</sup>

$$\sigma(\omega) = \frac{1}{k_{\rm B}T} \sum_{\alpha=1}^{2} \rho_{\alpha} q_{\alpha}^2 D_{\alpha}(\omega)$$
(32)

We note, however, that the frequency dependent ionic conductivity can, in general, be related to the Fourier transform of the ionic current-current time correlation function.38 Since the electric current is a collective dynamical quantity, its time correlation function comprises of a self-part that corresponds to a summation over the velocity autocorrelation functions of the ions and a cross part involving the sum of the correlation functions of the velocities of distinct ions. The generalized Nernst-Einstein relation (eq 32) includes only the self-part and the cross part is ignored. Although the importance of the cross part is smaller than the self-part, it may not be negligible at high ion concentrations. However, at low and moderate ion concentrations, the contribution of the cross part is expected to be much smaller than that of the self-part. For example, at 0.5 M concentration of aqueous NaCl solution, an analysis of the simulation data of self-diffusion coefficients and conductivity reveals that the cross term reduces the static conductivity by less than 5%.27 We also note that in the calculation of the static and dynamic ionic structure factors which appear in the final expressions of the friction and the conductivity, the solvent is treated as a dielectric continuum. At high concentrations, the molecular details of the ion-solvent and solvent-solvent correlations may become important. Because of these approximations, the present theory is limited to solutions of not too high concentration (less than 1 M solution). However, the theory has been found to give reliable results for the static conductance of strong electrolytes even up to 2 M concentration.34,35

As explained in the last section, the total friction acting on the tagged ion is decomposed into two parts. The first part is due to the microscopic interaction of the tagged ion with the surrounding ions and the second part originates from the hydrodynamic coupling of the velocity of the tagged ion with the current modes of the surrounding particles. The time dependent microscopic electrolyte friction,  $\delta \zeta_{s,mic}(t)$  is given by eq 6 of section 3. The frequency dependence of the hydrodynamic (or the electrophoretic part) is obtained by a generalization eq 13 to frequency domain. After the frequency dependence of the microscopic and hydrodynamic frictions are obtained, we calculate the frequency dependent diffusion coefficients of the ions by using the Generalized Einstein relation (eq 3) and then we obtain the frequency dependent conductivity from the generalized Nernst–Einstein relation (eq 32). **4.1. Derivation of Debye-Falkenhagen Form of Frequency Dependent Friction.** Here we identify the limiting conditions under which the present theory reduces to the well-known Debye–Falkenhagen expression (eq 30) of the frequency dependent electrolyte friction. We note that in Debye–Falkenhagen theory, the frequency dependence of only the ion atmosphere relaxation contribution is considered and that of the electrophoretic term is ignored. Accordingly, we consider only the microscopic electrolyte friction as given by eq 6.

The ions are assumed to be point ions as in the Debye– Huckel theory. The use of the DH solutions of the direct correlation function and the diffusive limit of ionic van Hove functions leads to the following expression of the time dependent friction

$$\zeta_{\text{s,mic}}(t) = \frac{q_{\text{s}}^2 \kappa_{\text{D}}^2}{3\epsilon} \left[ \frac{e^{-D\kappa_{\text{D}}^2 t}}{\sqrt{2\pi Dt}} + \kappa_{\text{D}} e^{D\kappa_{\text{D}}^2 t} \{\Phi(\sqrt{2\kappa_{\text{D}}^2 Dt}) - 1\} \right]$$
(33)

where  $\Phi(x)$  is the error function and it is assumed that all ions have the same diffusion coefficient *D*. We note that the above expression of time dependent electrolyte friction becomes identical with the one derived by de Leon et al.<sup>55</sup> when the diffusion coefficient *D* in the right-hand side of eq 33 is replaced by  $D^0$ , which is the value of the ion diffusion coefficient in the limit of infinite dilution.

The Fourier–Laplace transform of eq 33 can be carried out analytically to obtain the following expression of the frequency dependent friction

$$\xi_{\rm s;mic}(\omega) = \frac{q_{\rm s}^2 \kappa_{\rm D}}{6\epsilon D} \frac{1}{1 + \frac{1}{\sqrt{2}} [1 - i\omega/D\kappa_{\rm D}^2]^{1/2}}$$
(34)

The expression of the frequency dependent friction as given by eq 34 can be rewritten in the following form

$$\xi_{\rm s;mic}(\omega) = \xi_{\rm s;mic}(0) \frac{1 + \sqrt{q}}{1 + \sqrt{q}[1 - i\omega\tau_{\rm atm}]^{1/2}} \qquad (35)$$

where  $\tau_{atm}$  is defined in eq 31. Equation 35 is identical with the DF expression (eq 30). Thus, it is clear from the above analysis that the present microscopic theory reduces to the DF theory in the limit of low ion concentration when finite sizes of the ions are ignored, electrophoretic effects are not included, and the collective dynamics of the ion atmosphere relaxation is described by diffusive motion.

We note that the DF theory incorporates the cross dynamical coupling of ions at the level of ion atmosphere relaxation (or the ionic van Hove functions) but not at the level of ionic velocity or current relaxation. Thus, like the present theory, the DF expression also does not include the effects of the so-called current cross terms. The contribution of such cross correlations is expected to be vanishingly small in the limit of low ion concentration, and therefore, the DF expression is regarded as the correct limiting expression of the frequency dependent ion atmosphere friction in an electrolyte solution.

**4.2. Derivation of the Conductivity Expressions of Chandra, Wei, and Patey.** Chandra, Wei and Patey (CWP)<sup>26</sup> have recently derived analytical expressions of  $\sigma(\omega)$  by employing a non-Markovian equation of motion of the self-van Hove function of a tagged ion. By employing exactly known short and long time constraints upon the ionic self-van Hove function, these authors derived closed-form expressions of the frequency dependent conductivity for two models, referred to as models I

and II. Model I depends on an expression of the frequency dependent diffusion coefficient which ensures that the short time dynamics up to the second frequency moment and also the long time dynamics of the self-van Hove function are described correctly. The final expression of the conductivity is described by a multiple Debye form<sup>26</sup>

$$\sigma(\omega) = \frac{1}{k_{\rm B}T} \sum_{\alpha=1}^{2} \frac{\rho_{\alpha} q_{\alpha}^2 D_{\alpha}}{1 - i\omega D_{\alpha} m_{\alpha}/k_{\rm B}T}$$
(36)

where  $m_{\alpha}$  is the mass of an ion of species  $\alpha$ . The model II of CWP is a generalized version which correctly describes the short time dynamics up to the fourth frequency moment and also the long time dynamics of the ionic self-van Hove functions. In this model,  $\sigma(\omega)$  is given by<sup>26</sup>

$$\sigma(\omega) = \frac{1}{k_{\rm B}T} \sum_{\alpha=1}^{2} \rho_{\alpha} q_{\alpha}^{2} \left[ \frac{D_{\alpha} \Omega_{\alpha}^{2} - i\omega k_{\rm B} T/m_{\alpha}}{\Omega_{\alpha}^{2} - \omega^{2} - i\omega D_{\alpha} \Omega_{\alpha}^{2} m_{\alpha}/k_{\rm B} T} \right]$$
(37)

where  $\Omega_{\alpha}$  is the Einstein frequency of an ion of species  $\alpha$ , so  $\Omega_{\alpha}^2$  is proportional to the mean square force acting on an ion. The CWP theory is capable of predicting the increase of conductivity at low frequency. However, no attempt was made by CWP to derive the DF expression.

It is discussed here that the main expressions of CWP can be obtained from the present formalism. To recover the expression of Model I (eq 36), we ignore the frequency dependence of  $\zeta_s(\omega)$  in the Einstein relation and replace it by the zero-frequency value  $\zeta_s$  so that

$$D_{s}(\omega) = \frac{k_{\rm B}T}{m_{\rm s}} \frac{1}{-i\omega + \zeta_{\rm s}}$$
$$= \frac{D_{\rm s}}{1 - i\omega D_{\rm s} m_{\rm s}/k_{\rm B}T}$$
(38)

where, in deriving the second equality, we have used the Einstein relation  $\zeta_s = k_{\rm B}T/D_sm_s$ . On combining eq 38 with the Nernst–Einstein relation (eq 32), one obtains eq 36, which is the CWP model I expression of the frequency dependent conductivity. Thus, in this limiting case, the frequency dependence of the conductivity originates entirely from the inertial effects.

Also, when  $D_1 = D_2$  and  $m_1 = m_2$ , the frequency dependence of the conductivity is described by a simple Debye form

$$\sigma(\omega) = \frac{\sigma}{1 - \mathrm{i}\omega Dm/k_{\mathrm{B}}T} \tag{39}$$

where  $\sigma$  is the conductivity at zero frequency. We note that the simple Debye form has been used in the literature as an empirical expression for  $\sigma(\omega)$ .<sup>25</sup>

To derive the expression of model II (eq 37), we approximate the time dependence of the total friction as friction as

$$\zeta_{\rm s}(t) = \zeta_{\rm s}(t=0){\rm e}^{-t/\tau_{\rm s}} \tag{40}$$

where  $\zeta_s(t=0)$  is described by the spatial integral of the zerotime force-force correlation which is equal to the square of the Einstein frequency  $\Omega_s$  of the tagged ion. The relaxation time  $\tau_s$ can be eliminated in favor of the diffusion coefficient  $D_s$  by using the Einstein relation described above, and the resultant expression after Laplace transformation gives

$$\zeta_{\rm s}(\omega) = \frac{\Omega_{\rm s}^2}{-i\omega + D_{\rm s}m_{\rm s}\Omega_{\rm s}^2/k_{\rm B}T} \tag{41}$$

Equation 41, on combining with the generalized Einstein relation and eq 32, gives eq 37, which is the model II expression for  $\sigma(\omega)$ . Thus, in this model, the frequency dependence of conductivity originates from both the intertial and the non-Markovian effects.

### 5. Concentration Dependence of the Viscosity: Falkenhagen Expression

Viscosity of an electrolyte solution is long known to show anomalous dependence on ion concentration.<sup>7,56,57</sup> For some ions (like LiCl, NaCl, and BaCl<sub>2</sub> in water), the viscosity increases monotonically with concentration, while for some others (like KCl, KBr, or KI), the viscosity increases slightly at very low concentration, then it decreases as the ion concentration is increased and finally it increases again at higher concentration.<sup>7,58</sup> This anomalous concentration dependence of viscosity clearly has a complex origin which, to the best of our knowledge, has not even been addressed to from a microscopic theory.

At very low concentration, the increase in viscosity with ion concentration can be well described by the well-known expression derived by Falkenhagen many decades ago. The expression of Falkenhagen is valid for symmetrical electrolytes and is given by<sup>5,6,15</sup>

$$\eta(c) = \eta_0 + \frac{\kappa_{\rm D} \zeta_0}{480\pi} \tag{42}$$

where  $\eta(c)$  is the viscosity of the electrolyte solution when the molar concentration of the salt is c,  $\eta_0$  is the same of the pure solvent,  $\zeta_0$  is the friction on an ion in the solution at infinite dilution, and  $\kappa_D$  is the inverse Debye screening length. The Falkenhagen expression was later extended to asymmetrical electrolytes by Onsager and Fuoss.<sup>16</sup> Note, however, that at the heart of the derivation of Falkenhagen and also of Onsager and Fuoss is the Debye–Huckel theory of ion atmosphere. Since the Debye–Huckel theory is valid at very low concentrations, so is the Falkenhagen–Onsager–Fuoss expression. In fact, the validity of Falkenhagen expression seems to be limited to *even lower* concentrations than that of the DHO law of conductance for reasons we discuss below.

**5.1. Mode Coupling Theory Formulation for Viscosity.** Our starting point for the calculation of viscosity of the solution is the following Green–Kubo formula<sup>39</sup>

$$\eta = \lim_{k \to 0} \frac{1}{k_{\rm B} T V} \int_0^\infty \mathrm{d}t \, \langle \sigma^{zx}(k,0) \sigma^{zx}(k,t) \rangle \tag{43}$$

where  $\sigma^{zx}(k,t)$  is the transverse (or off-diagonal) component of the wavevector (**k**) and time dependent stress tensor. Clearly,  $\sigma^{zx}(k,t) = \sigma^{xz}(k,t)$ . Moreover, because of the inherent isotropy of the solution, the time correlation function  $\langle \sigma^{zx}(k,0)\sigma^{zx}(k,t)\rangle$  is independent of the particular combination (zx) of the Cartesian indices, it can equivalently be replaced by (xy) and (yz)components. The total transverse stress tensor of an ionic solution contains contributions from both solvent and ions and it can be written as

$$\sigma^{zx}(k,t) = \sigma^{zx}_{\text{sol}}(k,t) + \sigma^{zx}_{\text{ion}}(k,t)$$
(44)

where  $\sigma_{sol}^{zx}$  and  $\sigma_{ion}^{zx}$  denote, respectively, the solvent and ion contributions to the transverse stress. Accordingly, the viscosity can be written as

$$\eta = \eta_{\rm sol} + \eta_{\rm ion-sol} + \eta_{\rm ion} \tag{45}$$

where the terms in the right-hand side of the above equation represent, respectively, the solvent, ion-solvent, and ion contributions to the viscosity. At low concentration, the solvent contribution  $\eta_{sol}$  is expected to be very close to the to the viscosity of the pure solvent  $\eta_0$ . Since we are interested in dilute solutions in the present study, we assume that  $\eta_{sol}$  is equal to  $\eta_0$ . The ion-solvent contribution involves dynamical cross correlation between the transverse ion stress and the transverse solvent stress. The magnitude of such dynamical cross correlation is expected to be small for dilute solutions. Thus, the major ion dependent contribution comes from  $\eta_{ion}$ . In the present work, we calculate this ionic contribution, i.e., the one arising from ion-ion transverse stress correlation

$$\eta_{\text{ion}} = \lim_{k \to 0} \frac{1}{k_{\text{B}} T V} \int_{0}^{\infty} dt \left\langle \sigma_{\text{ion}}^{zx}(k,0) \sigma_{\text{ion}}^{zx}(k,t) \right\rangle$$
$$= \lim_{k \to 0} \frac{m^{2}}{k_{\text{B}} T V k^{2}} \int_{0}^{\infty} dt \left\langle \dot{J}_{x}(k,0) \dot{J}_{x}(k,t) \right\rangle$$
(46)

where  $J_x(k,t)$  is the time derivative of the transverse ion current. It is assumed that the wavevector k is along the *z*-axis and that all ions of the solution have the same mass *m*. We note that the ionic contribution to the viscosity was identified as the excess viscosity in earlier studies<sup>15,16</sup> although, strictly speaking, it becomes the excess viscosity only in the limit of very low ion concentration where solvent—ion cross dynamical stress correlation becomes vanishingly small. In the following, we employ the mode coupling treatment of Geszti<sup>59</sup> to derive a simplified microscopic expression of this ionic contribution  $\eta_{ion}$ .

Since we assume that the ionic contribution to viscosity arises only from the interaction between the ions and that the solvent modes relax much faster than the ionic modes in dilute solutions (as discussed before), one can use the collective ionic variables to describe the dynamics of the correlation functions of eq 46. We treat the mode-mode coupling among four variables of ionic current density  $j_{\alpha}(\mathbf{k})$  ( $\alpha = 1, 2, 3$ ) and the charge density  $\rho_{c}(\mathbf{k})$ . Clearly,  $\rho_{c}(\mathbf{k}) = q_{1}\rho_{1}(\mathbf{k}) + q_{2}\rho_{2}(\mathbf{k})$ . The Green-Kubo formula of eq 46 can now be rewrirren as

$$\eta_{\text{ion}} = \lim_{\epsilon \to 0} \lim_{k \to 0} \frac{m^2}{k^2 V} \int_0^\infty dt \, (QLj_1(\mathbf{k})|\exp(iQLQt - \epsilon t)|QLj_1(\mathbf{k})|$$
(47)

where *L* is the same Liouville operator which appeared in eq 12 and *Q* is the projector onto the manifold of dynamical variables orthogonal to ionic current density and the charge density. The time evolution operator  $\exp[iQLQt - \epsilon t]$  acts on the space of dynamical variables. The standard approximation in the mode-mode coupling expansion is to consider the subspace of various binary products of the basic variables. Among such binary products, the odd ones with respect to time inversion do not contribute to the viscosity; thus only the even combinations need be retained. Of these, the most important contribution comes from the charge density combination  $\rho_c \rho_c$ because this term decays slowly, particularly at low concentration when the Debye length ( $\lambda_D$ ) is large (because the ion atmosphere relaxation time is given by $\lambda_D^2/D_{ion}$ ). The contribution of the current combination *jj* is expected to be small because the current correlation function decays much faster (its relaxation time being determined by  $m/\zeta$ ). The *jj* term is, therefore, neglected in this work and we consider the contribution from the  $\rho_c \rho_c$  term.

The final expression of the excess viscosity is given by<sup>37</sup>

$$\eta_{\rm ex} = \frac{k_{\rm B}T}{60\pi^2} \int_0^\infty \mathrm{d}k \int_0^\infty \mathrm{d}t k^4 S_{\rm c}'(k)^2 F_{\rm c}(k,t)^2 / S_{\rm c}(k)^4 \quad (48)$$

where  $S_c(k)$  and  $F_c(k,t)$  are, respectively, the static and dynamic charge density structure factors of the ions in the solution.  $S'_c(k)$  represents the derivative of  $S_c(k)$  with respect to the wavevector k. Note that this expression has the same form as the one derived by Geszti,<sup>59</sup> except that the stucture factor here refers to that of charge density—not atomic and molecular density.  $F_c(k,t)$  thus describes the relaxation of ion atmosphere at all length scales.

If one includes the current-current contribution to the viscosity, an additional term enters in eq 48. The expression is similar, except  $F_c(k,t)$  gets replaced by the current-current correlation function, with a few additional changes. As alreday discussed, the contribution of this term to viscosity is small and is usually neglected. It should be pointed out here that an expression like eq 48 has been used to address the observed anomalous rise of viscosity of polyelectrolyte solutions at very low polyelectrolyte concentrations.<sup>60,61</sup> In this case, one needs to consider both the mass and charge density of the polyelectrolytes as the slow variables.

**5.2. Reduction to the Falkenhagen Expression.** In this section we show that the present theory of the ionic viscosity nicely reduces to the Falkenhagen expression<sup>5,6,15</sup> for a binary symmetric electrolyte in the limit of very low ion concentration. At low concentration, we use the Debye–Huckel expressions of the ionic pair correlation (which is an exponentially decaying function of the distance) and the direct correlation functions. The dynamic charge structure factor  $F_c(k,t)$  is determined in the following way. We assume that the dynamics of ion atmosphere relaxation occurs by diffusive motion; thus, we ignore the frequency dependence of  $D(\omega)$ ) in eq 8. The inverse Laplace transformation of eq 8 then gives the following dynamical equation for  $F_c(k,t)^{37}$ 

$$\frac{\partial F_{\rm c}(k,t)}{\partial t} = -Dk^2 [1 - 2\rho_1 c_{11}(k)] F_{\rm c}(k,t)$$
(49)

so that

$$\int_{0}^{\infty} [F_{\rm c}(k,t)/S_{\rm c}(k)]^2 \, \mathrm{d}t = \frac{1}{2D(k^2 + \kappa_{\rm D}^2)} \tag{50}$$

where, in deriving the above equation, we have used the Debye– Huckel expression of the direct correlation function. We now substitute eqs 50 in eq 48 and use the Debye–Huckel solutions of the ionic structure factors to obtain

$$\eta_{\rm ex} = \frac{k_{\rm B} T \kappa_{\rm D}^4}{30\pi^2 D} \int_0^\infty {\rm d}k \, \frac{k^2}{(k^2 + \kappa_{\rm D}^2)^3} \\ = \frac{\kappa_{\rm D} \zeta}{480\pi}$$
(51)

where the Einstein relation between the diffusion coefficient and the friction is used in deriving the second equality. Equation 51 becomes identical to the Falkenhagen expression of excess



**Figure 1.** The ion atmosphere contribution to the microscopic friction is plotted against the square root of ion concentration (in molarity) for a 1:1 electrolyte. The values of different parameters describing the solution are described in the text. The details of the calculations are available in ref 35. The solid curve shows the results of the present theory and the dashed curve represents the predictions of DHO theory. (Reprinted with permission from ref 35. Copyright 1999 American Institute of Physics.)

viscosity when the ionic friction is replaced by its value at infinite dilution. The same expression was also derived by Fuoss and Onsager.<sup>16</sup>

# 6. Numerical Results at Finite Concentration: Departure from Classical Laws

The numerical results of the electrolyte friction, conductivity and the viscosity are discussed for symmetric binary salt solutions at varying concentrations. All ions of the solutions are assumed to be charged hard spheres of equal diameter  $\sigma$ for simplicity. The solutions are completely specified by specifying the reduced charge  $q_1^* = \sqrt{q_1^2/k_{\rm B}T\sigma}$ , the reduced ion density  $\rho_1^* = \rho_1\sigma^3$  and the dielectric constant  $\epsilon$ . The pair correlation functions for the evaluation of the quantities  $S_{\alpha\beta}(k)$ and  $c_{s\alpha}(k)$  are obtained from the solutions of Attard.<sup>62</sup> In this scheme, the functional forms of the ionic pair correlation functions are the same as given by Debye–Huckel theory. However, the screening parameter which enters into the mathematical expressions is *not* the one of Debye–Huckel theory but a renormalized one. In real space, the expression of the pair correlation function  $h_{\alpha\beta}(r)$  is given by

$$h_{\alpha\beta}(r) = -\frac{q_{\alpha}q_{\beta}}{\epsilon k_{\rm B}T(1+\kappa\sigma)}\frac{\kappa^2}{\kappa_{\rm D}^2}\frac{e^{-\kappa(r-\sigma)}}{r}$$
(52)

where the screening parameter  $\kappa$  is related to the Debye screening parameter  $\kappa_D$  by the following relation

$$\kappa = \frac{\kappa_{\rm D}}{\left[1 - (\kappa_{\rm D}\sigma)^2/2 + (\kappa_{\rm D}\sigma)^3/6\right]^{1/2}}$$
(53)

The above solution of the ionic pair correlation function with the renormalized screening length has been found to be accurate up to 1 M concentration for monovalent electrolytes and it considerably extends the range of validity of the classical Debye–Huckel theory. The Fourier transform of the pair correlation function can be obtained readily from eq  $52.3^{5}$ 

**6.1. Static Electrolyte Friction and Conductance.** The numerical results of the electrolyte friction and conductance as predicted by the present microscopic theory are described in Figures 1-4.<sup>63</sup> The results of the ion atmosphere contribution to the electrolyte friction of a tagged ion,  $\delta \zeta_{s,mic}$ , are shown in Figure 1. In these calculations (and also the ones of Figures 2-3 and 5-8), the value of the reduced charge  $q^*$  is 14.1, which corresponds to the charge of a univalent ion of diameter 2.82 Å at T = 298 K. The reduced ion density  $\rho_1^*$  is varied from 0.0



**Figure 2.** The ion atmosphere contribution to the hydrodynamic part of ion diffusion is plotted against the square root of ion concentration. The different curves are as in Figure 1. (Reprinted with permission from ref 35. Copyright 1999 American Institute of Physics.)



**Figure 3.** The total conductance is plotted against the square root of ion concentration.  $\Lambda_0$  represents the conductance at infinite dilution. The different curves are as in Figure 1. (Reprinted with permission from ref 35. Copyright 1999 American Institute of Physics.)

to 0.0135 which corresponds to changing the concentration from 0.0 to 1.0 M where M is the molarity of the solution. The results of DHO law are also included in the figure for comparison. The DHO law predicts a linear increase of  $\delta \zeta_{s,mic}$  with the square root of ion concentration whereas the present theory predicts a nonlinear dependence at finite ion concentration. Here by finite ion concentration we mean a concentration regime higher than 0.05 M or so. In the limit of very low ion concentration (<0.05 M), the results of the present theory coincides with those of DHO law, as expected. However, at finite concentration the electrolyte friction is found to be much smaller than the prediction of the DHO law. Also, it changes rather weakly with concentration compared to the prediction of the DHO law. The electrophoretic contribution to the diffusion of the tagged ion is shown in Figure 2. In this figure,  $\delta D_{s,hyd}$  is plotted against square root of ion concentration where  $\delta D_{s,hyd} = k_B T / \xi_{s,hyd}$   $k_{\rm B}T/\zeta_{\rm s,hvd}^0$ . The results of  $\delta D_{\rm s,hyd}$  are shown in units of  $k_{\rm B}T/\eta\sigma$ . Again, DHO results are shown for comparison. At finite concentration, the predictions of the present theory are seen to be significantly different from those of the DHO law.

In Figure 3, we have shown how the ion conductance changes with square root of ion concentration for the same model solution. The values of the conductance at different concentrations are normalized by its limiting value at zero concentration. The details of the calculations are described in ref 35. Significant departure from DHO linear behavior is observed at finite concentration where a much weaker and nonlinear dependence of the conductance on square root of ion concentration is predicted by the present microscopic theory. The relative contributions of the microscopic relaxation and electrophoretic effects to the conductance have also been computed.35 The two effects have been found to exhibit different concentration dependence at higher ion concentration. Also, the electrophoretic effect has been found to be more important than the microscopic relaxation effect at finite ion concentration. This is in agreement with the findings of Altenberger and Friedman<sup>19</sup> in their Smoluchowski level description of ion conductance. The results shown in Figures 1-3 are obtained by using Attrad's solutions<sup>62</sup> of the ionic pair correlation functions. To verify the accuracy



**Figure 4.** The total conductance of aqueous (a) KCl and (b) NaCl solutions is plotted against the square root of ion concentration. The details of the calculations are available in refs 34 and 35. The solid curve represents the predictions of the present theory and the squares represent the experimental results. (Reprinted with permission from ref 34. Copyright 1999 American Chemical Society.)

of these results, calculations have also been carried out by using hypernetted chain (HNC) approximation of the ionic pair correlations and the results of conductance with HNC solutions have been found to be remarkably close to the ones of Figure  $3.^{35}$ 

Finally we compare the predictions of the present theory with experimental results of real ionic solutions. Both KCl and NaCl solutions are chosen as examples. The details of the numerical calculations can be found in refs 34 and 35. The theoretical predictions for these solutions along with experimental results are shown in Figure 4. The experimental results are available in the textbook of Glasstone<sup>1</sup> and also in ref 19. It is clear from this figure that the theoretical predictions are in excellent agreement with the experimental results even at high concentrations. This is quite impressive given that the theory does not involve *any* adjustable parameter.

**6.2. Frequency Dependent Electrolyte Friction and Conductivity.** The electrolyte friction and the conductivity become complex quantities at finite frequencies. We decompose the frequency dependent microscopic electrolyte friction into its real and imaginary parts as follows

$$\delta \zeta_{\text{s;mic}}(\omega) = \delta \zeta'_{\text{s;mic}}(\omega) + i \delta \zeta''_{\text{s;mic}}(\omega)$$
(54)

Similar decomposition is also made for the electrophoretic friction and the conductivity functions for the discussion of numerical results at finite frequencies. The frequency dependence of the real and imaginary parts of the microscopic electrolyte friction have been calculated for 0.01 and 0.1 M solutions of a 1:1 electrolyte at room temperature.<sup>36</sup> The results for the 0.1 molar solution are shown in Figure 5. The values of the friction at different frequencies are normalized by its zerofrequency value and the reduced frequency  $\omega^* = \omega \sigma^2 / D$ . At low concentration and at low frequency, the results of the present theory have been found to be quite close to the DF results.<sup>36</sup> However, at high concentration, the results of the present theory show significant departure from the DF results, especially at high frequency. Also, with increase of ion concentration, the dispersion of the electrolyte friction is found to occur at a higher frequency because of faster relaxation of the ion atmosphere.<sup>36</sup>

In Figure 6, we have shown the frequency dependence of the electrophoretic term for the 0.1 M solution. The dispersion of the electrophoretic contribution is seen to occur at a much higher frequency as one would expect. In fact, any noticeable



**Figure 5.** The (a) real and (b) imaginary parts of the frequency dependent microscopic electrolyte friction on an ion are plotted against frequency for a 0.1 M solution of a 1:1 electrolyte. The details of the calculations are available in ref 36. The solid and the dashed curves represent, respectively, the results of the present theory and of Debye–Falkenhagen expression (eq 30). (Reprinted with permission from ref 36. Copyright 2000 American Institute of Physics.)



**Figure 6.** The electrophoretic friction on an ion is plotted against frequency for a 0.1 M solution of a 1:1 electrolyte. The solid and the dashed curves represent, respectively, the real and the imaginary parts. (Reprinted with permission from ref 36. Copyright 2000 American Institute of Physics.)

change in the electrophoretic term is found only above  $\omega^* = 10^3$  which corresponds to a frequency well above the GHz region. The dispersion of the solvent contribution to the ionic friction, which is assumed to be a constant in the present calculations, is also expected to occur in this frequency domain. The solvent contribution, being a larger effect, may dominate the dispersion of the total ionic friction at such high frequencies.

The results of the frequency dependence of real and imaginary parts of the ion conductivity are shown in Figure 7 for 0.01 and 0.1 M solutions. The real part shows a slight increase at low frequency and then it decreases at high frequency as one would expect. The initial increase at low frequency can be attributed to the Debye-Falkenhagen effect. It is seen that the primary dispersion of the conductivity occurs at a much higher frequency than that of the microscopic electrolyte friction. The dispersion of the microscopic electrolyte friction is primarily determined by the inverse relaxation time of the ion atmosphere. The frequency dependence of the conductivity, on the other hand, is determined by the frequency dependent diffusion coefficient  $D(\omega)$ . The primary dispersion of  $D(\omega)$  occurs at a much higher frequency than that of the electrolyte friction because of the presence of  $-i\omega$  term in the generalized Einstein relation which connects the diffusion coefficient to the friction at finite frequencies. Sometimes the experimental results of the frequency dependent conductivity are analyzed in terms of the so-called Cole-Cole plots where the imaginary part of the conductivity is plotted against the real part at different frequencies. Such Cole-Cole plots show non-Debye behavior at higher ion concentration.<sup>36</sup> This is not unexpected as the Debye form of  $\sigma(\omega)$  (eq 39) is valid only in limiting situations as discussed in



**Figure 7.** The (a) real and (b) imaginary parts of the frequency dependent conductivity are plotted against frequency for 0.01 M (dashed curves) and 0.1 M solutions (solid curves) of a 1:1 electrolyte. The details of the calculations are available in ref 36. (Reprinted with permission from ref 36. Copyright 2000 American Institute of Physics.)



**Figure 8.** The ionic contribution to the viscosity is plotted against the square root of ion concentration (in molarity) for solutions of (a) 1:1 and (b) 2:2 electrolytes. The details of the calculations are available in ref 37. The reduced viscosity  $\eta_{ion}^* = \eta_{ion}\sigma/\zeta_0$ . The solid curves show the results of the present theory and the dashed curves represent the predictions of Falkenhagen expression (eq 42). (Reprinted with permission from ref 36. Copyright 2000 American Institute of Physics.)

section4 and, in general, the dispersion of the conductivity shows a more complex behavior.

6.3. Ionic Contribution to the Viscosity. The results of the ionic contribution to the viscosity of a 1:1 salt solution at varying concentration are shown in Figure 8a. The values of various parameters are the same as in Figure 1. In this figure, the results of the Falkenhagen expression are also included for comparison. The Falkenhagen expression predicts a linear increase of  $\eta_{ion}$ with square root of ion concentration whereas the present theory predicts a nonlinear dependence at finite ion concentration. In the limit of very low ion concentration (<0.05 M), the results of the present theory coincides with those of Falkenhagen, as expected. However, at finite concentration the excess viscosity is found to be much higher than the prediction of the Falkenhagen expression. In Figure 8b, we present the results for a 2:2 electrolyte. Here  $q^* = 28.2$  and the values of other parameters are the same as in the previous figure. A much stronger dependence of the excess viscosity on ion concentration is found compared to the results of 1:1 electrolytes. Also, the deviations from the Falkenhagen limiting behavior are found to be much more in this case which can be attributed to the stronger ionion correlations.

The microscopic theory predicts a stronger increase of viscosity with increase of ion concentration, especially for ions of higher valence which is in qualitative agreement with experimental results. At finite concentration, the viscosity increases nonlinearly with square root of concentration<sup>37</sup> contrary to the linear increase predicted by the Falkenhagen expression. Although the present theory proves to be an improvement over the Falkenhagen expression, a quantitative comparison of the viscosity data7,58 show that for many electrolytes (like NaCl in water), the viscosity is still underestimated by the present theory at higher concentrations. Clearly, much more work is needed to fully understand the concentration dependence of the viscosity. We note that although the static and dynamic ion-ion correlations and the screening effects are included in the present work, the molecular details of the ionsolvent and solvent-solvent correlations are missing. It would be interesting to study how these ion-solvent and solventsolvent correlations contribute to the excess viscosity of an electrolyte solution. The main approximation in the present formalism seems to be the neglect of the binary terms involving the solvent density in the set of slow variables. In a generalized theory, not only the charge and size of the ions, but the molecular details of solvent could also be important. Second, we may need to include the contributions of charge currents. Although the charge current is not a conserved quantity, its contribution needs to be evaluated. However, it is believed that the binary terms involving the ion-solvent and solvent-solvent correlations hold the key to many of the surprising results of the concentration dependence of viscosity.

## 7. Toward a Full Microscopic Theory: A Future Problem

There is yet no first principles theory of electrolyte conductance and viscosity which treats the ions and the solvent molecules at the same level. This is what we refer to as a fully microscopic theory. Since the solvent molecules here are dipolar, one needs to include not only the position but also the orientation of these molecules. Not only the solvent—solvent but the ion dipole interactions are also orientation dependent. This is exactly the difficulty which has limited any progress toward a full theoretical description. Here we describe some of the progress made recently.

In the calculation of the total microscopic friction, we require an expression for the dielectric friction from ion—solvent interaction. This can be derived by using the density functional theory, along with Kirkwood's formula and is given by

$$\delta \xi_{\rm mic,solv}(t) = \frac{2k_{\rm B}T}{3(2\pi)^3} \sum_{\alpha=1}^2 \int d\mathbf{k} \, d\Omega k^2 c_{\rm sol}(k) \times \sqrt{\rho_{\alpha}\rho_3} G_{\alpha3}(k,\Omega,t) c_{\rm s3}(k,\Omega) F_{\rm s}(k,t) + \frac{k_{\rm B}T\rho_3}{3(2\pi)^3} \int d\mathbf{k} \, d\Omega \, d\Omega' \, k^2 c_{\rm s3}(k,\Omega) G_{33}(k,\Omega,\Omega',t) \times c_{\rm s2}(k,\Omega') F_{\rm s}(k,t)$$
(55)

where  $\rho_3$  is the density of the solvent (species 3),  $c_{s3}(k,\Omega)$  is the wavevector and angle ( $\Omega$ ) dependent direct correlation function between the tagged ion and the solvent, and  $G_{\alpha3}(k,\Omega,t)$ and  $G_{33}(k,\Omega,\Omega',t)$  are the ion–solvent and solvent–solvent van Hove functions, respectively. This expression is a generalization of the friction for limiting ionic conductivity<sup>45,46</sup> to finite concentration; the quantities  $c_{s3}(k,\Omega)$ ,  $G_{\alpha3}(k,\Omega,t)$  and  $G_{33}$ -  $(k,\Omega,\Omega',t)$  now depend on ion concentration. One can similarly derive an expression for the dielectric friction from solvent polarization current modes.

The calculation of the ion-solvent and solvent-solvent van Hove functions will require solutions of a coupled hydrodynamic equations of the type discussed by Bagchi and Chandra<sup>64</sup> for dipolar liquids, but now for a composite system with ions. This leads to a set of coupled equations which can be linearized with respect to density and momentum fluctuations. Recently, Mahajan and Chandra<sup>65</sup> have solved such linearized hydrodynamic equations to obtain the time dependence of various ionion, ion-solvent and solvent-solvent van Hove functions. The implementation of such generalized solutions for the calculation of electrolyte friction is yet to be carried out. The decrement of dielectric constant<sup>49,66–72</sup> with ion concentration is another issue which is still not fully understood and not explicitly included in the existing theories of electrolyte friction and conductance. This effect will enter naturally once the dynamical coupling with the solvent modes is treated explicitly. The detailed theory will no doubt be quite involved, but will certainly be highly rewarding.

#### 8. Summary and Conclusions

In this article a review is presented mainly of our recent work on the microscopic derivations of the limiting tranpsort laws of electrochemistry and their generalizations to higher concentrations. All the three transport laws have been derived from the mode coupling theory approach. These derivations required identification of the proper dynamical variables and the use of appropriate time correlation functions. Although the derivations are by no means trivial, we believe that they are more transparent and easily understandable than the original derivations. The present analysis also leads to a better understanding of the origin of the often unusual forms of these laws. For example, the derivation of Falkenhagen–Onsager–Fuoss form<sup>15,16</sup> for viscosity very clearly reveals the origin of the 1/480 $\pi$  factor in the expression of the excess viscosity.

Numerical calculations of the full expressions show much better agreement with experimental results than the limiting expressions at small but nonnegligible concentrations, like 0.5 M. In fact, the agreement for the static conductance is quite impressive. The study of excess viscosity seems to suggest that direct solvent involvement might be playing a crucial role, even at as low a concentration as 0.1 M.

While we propose the full molecular theory, where the dynamical coupling between the solvent and the ions are treated explicitly in the calculation of the friction and the viscosity, as the most important problem in this area, there are several approximations in the present formulation that require improvement. For example, no account of the collective effects arising from the velocity cross correlations between different ions have been considered. Such a correlation is zero at t = 0 and small at small times, but can be significant at intermediate times for higher ion concentrations, like 1 M. However, we feel that the approximation involved in neglecting the explicit ion—solvent coupling at molecular level will play a far bigger role at higher concentrations.

Acknowledgment. We thank Dr. Ranjit Biswas for collaboration in the initial part of this work. The research reported here was supported in part by grants from the Council of Scientific and Industrial Research (CSIR), Department of Science and Technology (DST), Government of India and Indian National Science Academy (INSA).

#### **References and Notes**

- (1) Glasstone, S. An Introduction to Electrochemistry; Litton Educational Publishing, Inc.: New York, 1942.
- (2) Harned, H. S. The Physical Chemistry of Electrolyte Solutions, Reinhold: New York, 1958.
- (3) Petrucci, S., Ed. *Ionic Interactions*, Vol. 1; Academic Press: New York, 1971; Vol. 1.
- (4) Resibois, P. M. V. *Electrolyte Theory: An Elementary Introduction to Microscopic Theory*; Harper and Row: New York, 1968.
- (5) Falkenhagen, H.; Kelbig, G. in *Modern Aspects of Electrochemistry: No.* 2; Bokris, J. O'M., Ed.; Butterworths: London, 1959.
- (6) Falkenhagen, H. *Theorie der Elektrolyte*; Hirzel: Leipzig, 1971.
  (7) Stokes, R. H.; Mills, R. *Viscosity of Electrolytes and Related Properties*; Pergamon Press: Oxford, 1965.
- (8) Koryta, J.; Dvorak, J. Principles of Electrochemistry; Wiley: New York, 1987.
- (9) Friedman, H. L. A Course in Statistical Mechanics; Prentice Hall: Englewood Cliffs, 1985.
- (10) Bockris, J. O'M.; Reddy, A. K. N. *Modern Electrochemistry*, 2nd ed.; Plenum Press: New York, 1998.
- (11) Barthel, J. M. G.; Krienke, H.; Kunz, W. Physical Chemistry of Electrolyte Solutions: Modern Aspects; Springer: New York, 1998.
  - (12) Debye, P.; Huckel, H. Z. Phys. 1924, 25, 49.
  - (13) Onsager, L. Z. Phys. 1926, 27, 388; 1927, 28, 277.
  - (14) Debye, P.; Falkenhagen, H. Z. Phys. 1928, 29, 121; 1928, 29, 401.
- (15) Falkenhagen, H. Z. Phys. **1931**, 32, 745. Falkenhagen, H.; Vernon, E. L. Z. Phys. **1932**, 33, 140.
  - (16) Onsager, L.; Fuoss, R. M. J. Phys. Chem. 1932, 36, 2689.
  - (17) Debye, P.; Huckel, H. Z. Phys. 1923, 24, 185; 1923, 24, 305.
  - (18) Friedman, H. L. Physica 1964, 30, 537.
  - (19) Altenberger, A. R.; Friedman, H. L. J. Chem. Phys. 1983, 78, 4162.
  - (20) Zhong, E. C.; Friedman, H. L. J. Phys. Chem. 1988, 92, 1685.

(21) Bernard, O.; Kunz, W.; Turq, P.; Blum, L. J. Phys. Chem. 1992, 96, 398.

- (22) Bernard, O.; Kunz, W.; Turq, P.; Blum, L. J. Phys. Chem. 1992, 96, 3833.
- (23) Turq, P.; Blum, L.; Bernard, O.; Kunz, W. J. Phys. Chem. 1995, 99, 822.
- (24) Allen, M. P.; Tildesley, D. J. Computer Simulation of Liquids; Clarendon: Oxford, 1987.
- (25) Caillol, J. M.; Levesque, D.; Weis, J. J. J. Chem. Phys. 1986, 85, 6645.
  - (26) Chandra, A.; Wei, D.; Patey, G. N. J. Chem. Phys. 1993, 99, 2083.
  - (27) Lyubartsev, A. P.; Laaksonen, A. J. Phys. Chem. **1996**, 100, 16410.
- (28) Loffler, G.; Schreiber, H.; Steinhauser, O. J. Chem. Phys. 1997, 107, 3135.
- (29) Chandra, A. Phys. Rev. Lett. 2000, 85, 768.
- (30) Trullas, J.; Giro, A.; Padro, J. A. J. Chem. Phys. 1990, 93, 5177.
- (31) Canales, M.; Sese, G. J. Chem. Phys. 1998, 109, 6004.
- (32) Jordat, M.; Bernard, O.; Turq, P.; Kneller, G. R. J. Chem. Phys. **1999**, *110*, 7993.
  - (33) Svishchev, I. M.; Kusalik, P. G. Phys. A 1993, 192, 628.
- (34) Chandra, A.; Biswas, R.; Bagchi, B. J. Am. Chem. Soc. 1999, 121, 4082.
  - (35) Chandra, A.; Bagchi, B. J. Chem. Phys. 1999, 110, 10024.
  - (36) Chandra, A.; Bagchi, B. J. Chem. Phys. 2000, 112, 1876.
  - (37) Chandra, A.; Bagchi, B. J. Chem. Phys. 2000, 113, 3226.
- (38) Hansen, J. P.; McDonald, I. R. *Theory of Simple Liquids*; Academic: London, 1986.
- (39) Balucani, U.; Zoppi, M. Dynamics of the Liquid State; Clarendon Press: Oxford, 1994.
- (40) Hynes, J. T.; Kapral, R.; Weinberg, M. J. Chem. Phys. 1979, 70, 1456.
  - (41) Bhattacharyya, S.; Bagchi, B. J. Chem. Phys. 1997, 106, 1757.
  - (42) Bagchi, B. J. Chem. Phys. 1998, 109, 3989.
  - (43) Kirkwood, J. G. J. Chem. Phys. 1946, 14, 180.
  - (44) Colonomos, P.; Wolynes, P. G. J. Chem. Phys. 1979, 71, 2644.
  - (45) Biswas, R.; Roy, S.; Bagchi, B. Phys. Rev. Lett. 1995, 75, 1098.
  - (46) Biswas, R.; Bagchi, B. J. Am. Chem. Soc. 1997, 119, 5946. Bagchi,
- B.; Biswas, R. Acc. Chem. Res. **1998**, 31, 181.
- (47) Sjogren, L.; Sjolander, A. J. Phys. C: Solid State Phys. 1979, 12, 4369.
  - (48) Gaskell, T.; Miller, S. J. Phys. C: Solid State Phys. 1978, 11, 3749.
  - (49) Hubbard, J. B.; Onsager, L. J. Chem. Phys. 1977, 67, 4850.
  - (50) Zwanzig, R. J. Chem. Phys. 1963, 38, 1603.

(51) Chandra, A.; Jana, D.; Bhattacharjee, S. J. Chem. Phys. 1996, 104, 8662.

(52) Onsager, L. Ann. NY Acad. Sci. 1945, 46, 241.

(53) The conductance  $(\Lambda_{\alpha})$  and the self-diffusion coefficient  $(D_{\alpha})$  of species  $\alpha$  are related by the equation:  $\Lambda_{\alpha} = q_{\alpha}FD_{\alpha}/k_{B}T$  where *F* is Faraday.

- (54) Anderson, J. E. J. Non-Cryst. Solids 1994, 172–174, 1190.
  (55) de Leon, G. C.; Medina-Nayola, M.; Alarcon-Waess, O.; Ruiz-Estrada, H. Chem. Phys. Lett. 1993, 207, 294.
- (56) Jones, G.; Dole, M. J. Am. Chem. Soc. **1929**, *51*, 2950.
- (57) Applebey, M. P. J. Chem. Soc. **1910**, *97*, 2000.
- (58) CRC Handbook of Chemistry and Physics; Weast, R. C., Ed.; CRC Press: Boca Raton, FL, 1989.
  - (59) Geszti, T. J. Phys. C **1983**, 16, 5805.
- (60) Forster, S.; Schmidt, M. Advances in Polymer Science; Springer-Verlag: New York, 1995; Vol. 120, p 51.
  - (61) Yethiraj, A.; Bagchi, B. To be published.
- (62) Attard, P. Phys. Rev. E 1993, 48, 3604.

(63) Figures 1-3 are taken from ref 35 and Figure 4 is based on the results of ref 34. Figures 5-7 are taken from ref 36 and Figure 8 is from ref 37.

- (64) Bagchi, B.; Chandra, A. Adv. Chem. Phys. 1991, 80, 1.
- (65) Mahajan, K.; Chandra, A. J. Chem. Phys. 1997, 106, 2360.
- (66) Hubbard, J. B. J. Chem. Phys. 1978, 68, 1649.
- (67) Hubbard, J. B.; Colonomos, P.; Wolynes, P. G. J. Chem. Phys. 1979, 71, 2652.
  - (68) Winsor, P.; Cole, R. H. J. Phys. Chem. 1982, 86, 2486.
  - (69) Patey, G. N.; Carnie, S. L. J. Chem. Phys. 1983, 78, 5183.
  - (70) Kusalik, P. G.; Patey, G. N. J. Chem. Phys. 1988, 88, 7715.
  - (71) Chandra, A.; Wei, D.; Patey, G. N. J. Chem. Phys. 1993, 98, 4959.
  - (72) Chandra, A. J. Chem. Phys. 2000, 113, 903.